Author's response to reviews

Title: A Longitudinal Study of Maternal Prenatal Quality of Life: Can It Predict Preterm Birth and Low Birthweight?

Authors:

Panchalli Wang (03461@cych.org.tw)
Shwu-Ru Liou (srliou5022@gmail.com)
Ching-Yu Cheng (chingyuus@gmail.com)

Version: 2 Date: 7 September 2012

Author's response to reviews: see over
Emily Crow  
Executive Editor of the BMC Pregnancy and Childbirth

Dear Dr. Emily Crow, the Executive Editor:

We appreciate your giving us a chance to revise our manuscript titled “A Longitudinal Study of Maternal Prenatal Quality of Life: Can It Predict Preterm Birth and Low Birthweight?” (MS# 1613625693791606). In this revision, we added a paragraph of Ethical Considerations in the Methods section as suggested. We appreciate your kindly consideration and look forward to hearing from you again.

Sincerely yours,

Panchalli Wang, MD  
Chia-Yi Christian Hospital, Taiwan

Shwu-Ru Liou, PhD, RN  
Associate Professor  
Chang Gung University of Science and Technology, Taiwan

Ching-Yu Cheng, PhD, RN  
Associate Professor  
Chang Gung University of Science and Technology, Taiwan